Author's response to reviews

Title: A population-based study of the extent of colorectal cancer screening in men with HIV

Authors:

Tony Antoniou (tantoniou@smh.ca)
Nathaniel Jembere (Nathaniel.jembere@ices.on.ca)
Refik Saskin (refik.saskin@ices.on.ca)
Alexander Kopp (alexander.kopp@ices.on.ca)
Richard H. Glazier (rick.glazier@ices.on.ca)

Version: 2 Date: 28 May 2014

Author's response to reviews:

1. Changed 'Objectives' in Abstract to 'Background'.
2. Changed 'Introduction' heading in main body to 'Background'
3. Added Keywords on a separate page following abstract.
4. Added 'Conclusions' heading to main body of manuscript
5. Added List of abbreviations used section following Conclusions.
6. Organized 'Competing Interests', 'Authors' Contributions' and 'Acknowledgements' sections as recommended.
7. Moved Figure Legends prior to Tables.
8. Note that Results and discussion kept as two separate sections (seems to be okay to do this according to Instructions to Authors), and no Authors' information or Endnotes sections required.